Author's response to reviews

Title: Unit based perinatal mortality audit in all 15 perinatal co-operation units in the northern region of the Netherlands.

Authors:

Mariet Th. van Diem (m.t.van.diem@og.umcg.nl)
Albertus Timmer (a.timmer@umcg.nl)
Klasien A. Bergman (k.a.bergman@umcg.nl)
Katelijne Bouman (k.bouman@umcg.nl)
Nico van Egmond (n.vanegmond@hotmail.com)
Dennis A. Stant (a.d.stant@umcg.nl)
Lida H.M. Ulkeman (a.h.m.ulkeman@umcg.nl)
Wenda B. Veen (w.b.veen@umcg.nl)
Jan Jaap H.M. Erwich (j.j.h.m.erwich@umcg.nl)

Version: 2 Date: 31 December 2011

Author's response to reviews:

Dear Sir/Madam,

Thank you for your comments.

We have now provided a title page with names, institutions, countries and e-mail addresses of all authors and the full postal address of the corresponding author.

The abstract was included in the manuscript file. This abstract is identical to the one on the submission system.

Our comment on ethics approval is added to the methods section of the manuscript.

We hope that the manuscript is now suitable for starting the peer review process.

Yours sincerely,

Mariet van Diem